Author's response to reviews

Title: Comparative benefit from small tumour size and adjuvant chemotherapy: clues for explaining breast cancer mortality decline

Authors:

Romano Demicheli (Romano.Demicheli@istitutotumori.mi.it)
Federico Ambrogi (federico.ambrogi@unimi.it)

Version: 9 Date: 26 August 2014

Author's response to reviews: see over
Dear Dr. Thomas Ruhstaller

Journal Editorial Office - BioMed Central

The manuscript **Comparative benefit from small tumour size and adjuvant chemotherapy: clues for explaining breast cancer mortality decline** has been revised according to the point raised the Editor.

Regarding the Consent Statement, the sentence “The protocol design was approved by both the Scientific Committee and the Ethic Committee of the Istituto Nazionale Tumori and informed consent was requested to all patients” was changed with “The protocol design was approved by both the Scientific Committee and the Ethic Committee of the Istituto Nazionale Tumori and informed consent was obtained from all patients.”.

Best regards.

Romano Demicheli.

Federico Ambrogi.